Author’s response to reviews

Title: Food consumption and undernutrition variations among mothers during the post-harvest and lean seasons in Amoron'i Mania Region, Madagascar

Authors:
Lantonirina RAVAORISOA (lantonirinadr@yahoo.fr)
Julio Rakotonirina (juliorakotonirina@yahoo.fr)
Lalhyss Randriamanantsaina (rlalhyss05@gmail.com)
Jean de Dieu Marie Rakotomanga (rktjdm@yahoo.fr)
Michèle Dramaix (michele.dramaix@ulb.ac.be)
Philippe Donnen (pdonnen@ulb.ac.be)

Version: 5 Date: 12 Jul 2019

Author’s response to reviews:

We took into account all the comments of the editor and the reviewer:

- A summarizing statement explaining that methodology has been published previously: added
- The role of the funding body: added in the funding section.
- Funding information in the acknowledgements: removed.
- Test statistic for the p-value: added.
- Table 2: only variables that were included in the multivariable model.